Author's response to reviews

Title: Expression of CD80 and CD86 costimulatory molecules are potential markers for better survival in nasopharyngeal carcinoma

Authors:

Cheng-Shyong Chang (15120@cch.org.tw)
Julia H Chang (67842@cch.org.tw)
Nicholas C Hsu (99054@cch.org.tw)
Hsuan-Yu Lin (93687@cch.org.tw)
Chih-Yuan Chung (45413@cch.org.tw)

Version: 5 Date: 22 May 2007

Author's response to reviews: see over
Dear Editor,

My colleagues and I have revised the manuscript to facilitate the online production process. The revisions are listed as following:

1. The header is removed.
2. The affiliations are listed on separate lines.
3. "List of abbreviations" is changed to "Abbreviations".
4. A line indicating that all authors read and approved the final manuscript has been added in Authors' contributions.
5. Figure legends are placed before the tables, after the references.
6. The grey shading are removed from table 1. Bold text are used to highlight.

Thanks for accepting our manuscript!

Sincerely,

Cheng-Shyong Chang, M.D.
Division of Hematology-Oncology, Department of Internal Medicine, Changhua Christian Hospital, 135 Nanhsiao Street, Changhua 500, Taiwan.
Phone: +886 4 7238595  Ext. 1013
Fax:   +886 4 7232942